Author's response to reviews

Title: Recognizing Acute Delirium As part of your Routine [RADAR]: A validation study

Authors:

Philippe Voyer (philippe.voyer@fsi.ulaval.ca)
Nathalie Champoux (nathalie.champoux@umontreal.ca)
Johanne Desrosiers (johanne.desrosiers@usherbrooke.ca)
Philippe Landreville (philippe.landreville@psy.ulaval.ca)
Jane McCusker (jane.mccusker@mcgill.ca)
Johanne Monette (johanne.monette@mcgill.ca)
Maryse Savoie (maryse.savoie@vac-acc.gc.ca)
Sylvie Richard (sylvie.richard@fsi.ulaval.ca)
Pierre-Hugues Carmichael (pierre-hugues.carmichael.cha@ssss.gouv.qc.ca)

Version: 2 Date: 6 November 2014

Author's response to reviews:

For more than 20 years, detection of delirium is a challenge not yet overcome by all health care providers. In our validation study, we have addressed all barriers identified in previous studies. We have created the RADAR tool that takes only 7 seconds to administer and reaches a sensitivity beyond the available tools when use by clinical staff.